Author’s response to reviews

Title: Non-Structural Misalignments of Body Posture in the Sagittal Plane

Authors:

Dariusz Czaprowski (dariusz.czaprowski@interia.pl)
Lukasz Stolinski (stolinskilukasz@op.pl)
Marcin Tyrakowski (marcintyrak@gmail.com)
Mateusz Kozinoga (kozinoga@ump.edu.pl)
Tomasz Kotwicki (kotwicki@ump.edu.pl)

Version: 2 Date: 17 Dec 2017

Author’s response to reviews:

Dear Professor Theodoros B. Grivas, MD, PhD

Editor-In-Chief

Scoliosis and Spinal Disorders

First of all we would like to thank for positive comments and assessment of changes made in revised version of our manuscript. In order to facilitate the assessment of the changes made in the revised version of the manuscript, they have been highlighted in grey.

According to the Reviewers suggestion According to the text was carefully considered and improved in respect of grammatical errors.

When the revised version of the manuscript was uploaded, we attached only the Figures which was corrected. Therefore, we hope that after submitting this version of the paper everything will be fine with the Figures.

We are appreciate for all comments and suggestions of the Reviewers which allowed us to correct paper and improve its quality.

Yours sincerely,

Dariusz Czaprowski, PT, PhD, SOSORT
Department of Physiotherapy
Józef Rusiecki University College
Bydgoska 33
10-243 Olsztyn, Poland
e-mail: dariusz.czaprowski@interia.pl

Mateusz Kozinoga, PT, MSc, SOSORT
Spine Disorders and Pediatric Orthopedics Department, University of Medical Sciences, Poznan, Poland
28 Czerwca 1956 135/147 street
61-545 Poznan, Poland
e-mail: kozinoga@ump.edu.pl